Author’s response to reviews

Title: Pulmonary Tuberculous: Symptoms, diagnosis and treatment. 19-year experience in a third level pediatric hospital

Authors:

Napoleon Gonzalez Saldaña (nagosal@hotmail.com)
Marte Hernandez Porras (martehdzinp@yahoo.com.mx)
Patricia Saltigeral (psaltigeral@yahoo.com)
Victor Gomez (vgomez@hotmail.com)
Hugo Juarez Olguin (juarezol@yahoo.com)

Version: 2 Date: 2 August 2013

Author’s response to reviews: see over
Dear Editors

We are submitting to you as editors of BMC INFECTIOUS DISEASES, the original article entitled: “Pulmonary Tuberculous: Symptoms, diagnosis and treatment. 19-year experience in a third level pediatric hospital”, with the purpose that can be published in your prestigious Journal. We transfer all copyright ownership of the manuscript to BioMed Central in the event that the work is published. Authors warrant that article is original; is not under consideration by another publication. Authors confirm that final manuscript has been read and each author’s contribution has been approved by the appropriate authors. The authors have not published or submitted none related papers from the same study. The study was approved by Ethics Committee of this Institution. The authors declare that they have no competing interests.

Looking forward to your answer we remain yours truly.

Sincerely

Hugo Juárez Olguín
Correspondent author

Correspondence and reprints:
Hugo Juárez Olguín
Laboratorio de Farmacología, Instituto Nacional de Pediatría.
Avenida Imán N° 1, 3rd piso Colonia Cuicuilco CP 04530, Mexico City, MEXICO
Tel & Fax 5255 1084 3883 E-mail: juarezol@yahoo.com